Author's response to reviews

Title: Clinical and microbiological characteristics of bloodstream infections due to AmpC beta-lactamase producing Enterobacteriaceae: An active surveillance cohort in a large centralized Canadian region.

Authors:

Vikas P. Chaubey (vikas.chaubey@albertahealthservices.ca)
Johann D.D. Pitout (dr.johann.pitout@cls.ab.ca)
Bruce Dalton (bruce.dalton@albertahealthservices.ca)
Daniel B. Gregson (Dan.Gregson@cls.ab.ca)
Terry Ross (terry.ross@cls.ab.ca)
Kevin B. Laupland (klaupland@gmail.com)

Version: 2 Date: 12 January 2014

Author's response to reviews: see over
Re: Manuscript titled: “Clinical and microbiological characteristics of bloodstream infections due to AmpC β-lactamase producing Enterobacteriaceae: An active surveillance cohort in a large centralized Canadian region.”

December 21, 2013

Dear Dr. Harris;

Please find attached the above-captioned manuscript to be considered for publication in *BMC Infectious Diseases*. This is the largest study of its kind describing the clinical outcome and antimicrobial therapy of bacteremic patients infected with a broad range of AmpC beta-lactamase producing Enterobacteriaceae.

Thank you for considering our manuscript for publication. We look forward to your decision.

Sincerely,

Vikas P. Chaubey, MDCM, FRCPC